\begin{table}%tA.1 
%\centering !!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!
 \caption{\label{coresed} Core spectral energy distribution data.} 
 \begin{tabular}{llrrc}
 \hline \hline
  Filter & $\lambda$ & Flux & Obs. Date & Ref.\\
  &($\mu$m)&(mJy)&(yr, m, d)& \\
  \hline
4C~12.03      & & & & \\ 
VISIR/SiC    & 1.3e+01 & $<$3.1e+00 & 2006, 06, 18   & $ $           \\ 
5~GHz         & 6.0e+04 &     7.4e+00 & 1989, 03      & 24$^{*}$     \\ 
1.5~GHz       & 2.0e+05 &     4.0e+00 & 1989, 03      & 24           \\ 
3C~015        & & & & \\ 
 {\it F702W}        & 7.0e-01 & $<$7.6e-03 & $<$1999, 04   & 8            \\ 
VISIR/SiC    & 1.3e+01 & $<$3.3e+00 & 2006, 06, 18   & $ $           \\ 
22.4~GHz      & 1.3e+04 &     3.7e+01 & 2006, 08      & 12           \\ 
22~GHz        & 1.4e+04 &     1.8e+02 & 2002, 03      & 32           \\ 
18.5~GHz      & 1.6e+04 &     2.1e+02 & 2002, 03      & 32           \\ 
14.9~GHz      & 2.0e+04 &     3.2e+01 & 2006, 08      & 12           \\ 
8.35~GHz      & 3.6e+04 &     2.8e+01 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     3.0e+02 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     2.4e+01 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     7.6e+02 & 1993, 03      & 2            \\ 
3C~029        & & & & \\ 
Rosat/PSPC   & 1.2e-03 & $<$4.1e-05 & 1992         & 36           \\ 
F25SRF2      & 2.5e-01 &     1.1e-03 & 2000, 06, 08   & 9            \\ 
{\it F702W}        & 7.0e-01 &     9.5e-03 & 1995, 01, 12   & 7            \\ 
{\it F160W}        & 1.6e+00 & $<$7.1e-02 & 2004, 12, 04   & 26           \\ 
VISIR/SiC    & 1.3e+01 & $<$3.8e+00 & 2006, 12, 27   & $ $           \\ 
22~GHz        & 1.4e+04 &     9.1e+01 & 2002, 3       & 32           \\ 
18.5~GHz      & 1.6e+04 &     1.1e+02 & 2002, 3       & 32           \\ 
5~GHz         & 6.0e+04 &     9.3e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     5.5e+01 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     4.0e+01 & 1993--2003    & 2            \\ 
3C~040        & & & & \\ 
VISIR/SiC    & 1.3e+01 & $<$3.8e+00 & 2006, 12, 27   & $ $           \\ 
22~GHz        & 1.4e+04 &     1.5e+02 & 2002, 3       & 32           \\ 
18.5~GHz      & 1.6e+04 &     1.6e+02 & 2002, 3       & 32           \\ 
5~GHz         & 6.0e+04 &     1.0e+02 & 1989         & 29           \\ 
4.85~GHz      & 6.2e+04 &     9.0e+01 & 1984         & 21           \\ 
2.3~GHz       & 1.3e+05 &     7.9e+01 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     4.9e+01 & 93, 03        & 2            \\ 
3C~078        & & & & \\ 
Chandra      & 4.5e-04 &     1.3e-04 & $<$2005, 01   & 1            \\ 
BeppoSax     & 1.2e-03 &     3.0e-04 & 1997, 01, 07   & 35           \\ 
F25QTZ       & 2.5e-01 &     5.7e-02 & 2000, 03, 15   & 9            \\ 
{\it F555W}        & 5.6e-01 &     3.2e-01 & 1996, 09, 16   & 7            \\ 
{\it F702W}        & 7.0e-01 &     3.9e-01 & 1994, 08, 17   & 7            \\ 
F28X50LP     & 7.2e-01 &     6.6e-01 & 2000, 03, 15   & 9            \\ 
VISIR/SiC    & 1.3e+01 & $<$4.1e+00 & 2006, 12, 27   & $ $           \\ 
345~GHz       & 8.7e+02 &     2.8e+02 & 2001, 02, 11   & 31           \\ 
15~GHz        & 2.0e+04 &     6.9e+02 & 1982, 06, 18   & 33           \\ 
5~GHz2        & 6.0e+04 &     6.3e+02 & 1982, 06, 18   & 33           \\ 
5~GHz         & 6.0e+04 &     9.6e+02 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     5.8e+02 & 1994         & 30           \\ 
1.5~GHz       & 2.0e+05 &     7.5e+02 & 1982, 06, 18   & 33           \\ 
Fornax~A     & & & & \\ 
{\it F175W}        & 1.7e-01 &     6.3e-03 & 1992, 02, 01   & 14           \\ 
{\it F480LP}       & 5.1e-01 & $<$7.9e-03 & 1992, 02, 02   & 14           \\ 
{\it F555W}        & 5.6e-01 &     4.1e-02 & 1996--2006    & 25           \\ 
VISIR/SiC    & 1.3e+01 & $<$9.3e+00 & 2006, 12, 27   & $ $           \\ 
5~GHz         & 6.0e+04 &     2.6e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 & $<$4.0e+00 & 1994         & 30           \\ 
3C~093        & & & & \\ 
VISIR/SiC    & 1.3e+01 & $<$5.3e+00 & 2006, 12, 27   & $ $           \\ 
8.4~GHz       & 3.6e+04 &     3.9e+00 & 1991, 12      & 4            \\ 
5~GHz         & 6.0e+04 &     3.8e+00 & 1991, 12      & 4$^{*}$      \\ 
\hline
3C~098        & & & & \\ 
ROSAT        & 1.2e-03 &     5.8e-03 & $<$1999      & 18           \\ 
{\it F702W}        & 7.0e-01 & $<$2.6e-03 & $<$1999, 04   & 8            \\ 
VISIR/SiC    & 1.3e+01 &     2.0e+01 & 2005, 12, 18   & $ $           \\ 
345~GHz       & 8.7e+02 & $<$2.4e+01 & 2001, 02, 11   & 31           \\ 
8.35~GHz      & 3.6e+04 &     6.1e+00 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     9.0e+00 & 1985, 01      & 16           \\ 
3C~105        & & & & \\ 
{\it F160W}        & 1.6e+00 & $<$1.9e-02 & 2004, 10, 26   & 26           \\ 
VISIR/SiC    & 1.3e+01 & $<$2.2e+00 & 2005, 12, 18   & $ $           \\ 
22.4~GHz      & 1.3e+04 &     2.1e+01 & 2006, 08      & 12           \\ 
14.9~GHz      & 2.0e+04 &     2.0e+01 & 2006, 08      & 12           \\ 
8.35~GHz      & 3.6e+04 &     1.9e+01 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     1.4e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     1.9e+01 & 1994         & 30           \\ 
3C~120        & & & & \\ 
VISIR/SiC    & 1.3e+01 &     2.4e+02 & 2006, 12, 27   & $ $           \\ 
5~GHz         & 6.0e+04 &     3.5e+03 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     2.7e+03 & 1994         & 30           \\ 
3C~135        & & & & \\ 
{\it F160W}        & 1.6e+00 &     1.2e-02 & 2005,04,08   & 26           \\ 
VISIR/SiC    & 1.3e+01 & $<$4.1e+00 & 2006, 12, 27   & $ $           \\ 
8.35~GHz      & 3.6e+04 &     1.0e+00 & 89, 91        & 17           \\ 
5~GHz         & 6.0e+04 &     4.6e+00 & $<$1993      & 37           \\ 
Pictor~A     & & & & \\ 
VISIR/SiC    & 1.3e+01 &     6.8e+01 & 2006, 12, 27   & $ $           \\ 
22~GHz        & 1.4e+04 &     1.1e+03 & 2002, 03      & 32           \\ 
18.5~GHz      & 1.6e+04 &     7.8e+02 & 2002, 03      & 32           \\ 
5~GHz         & 6.0e+04 &     1.0e+03 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     7.5e+02 & 1994         & 30           \\ 
3C~198        & & & & \\ 
{\it F160W}        & 1.6e+00 & $<$1.9e-02 & 2005, 05, 03   & 26           \\ 
F25SRF2      & 2.5e-01 &     2.3e-02 & 2000, 03,15   & 9            \\ 
{\it F702W}        & 7.0e-01 &     5.4e-01 & $<$1999, 4    & 8            \\ 
VISIR/SiC    & 1.3e+01 & $<$4.3e+00 & 2006, 12, 27   & $ $           \\ 
345~GHz       & 8.7e+02 & $<$2.0e+01 & 2001, 02, 11   & 31           \\ 
5~GHz         & 6.0e+04 &     2.0e+00 & 1993--2003    & 3$^{*}$      \\ 
1.4~GHz       & 2.1e+05 &     1.6e+00 & 1993--2003    & 2            \\ 
3C~227        & & & & \\ 
F25SRF2      & 2.5e-01 &     1.3e-01 & 2000, 01, 27   & 9            \\ 
{\it F702W}        & 7.0e-01 &     4.8e-01 & 1995, 05, 19   & 9            \\ 
{\it F160W}        & 1.6e+00 & $<$7.2e-02 & 2005, 03, 28   & 26           \\ 
VISIR/SiC    & 1.3e+01 &     2.9e+01 & 2006, 12, 27   & $ $           \\ 
345~GHz       & 8.7e+02 & $<$2.1e+01 & 2001, 02, 11   & 31           \\ 
22.4~GHz      & 1.3e+04 &     1.2e+01 & 2006, 08      & 12           \\ 
14.9~GHz      & 2.0e+04 &     1.4e+01 & 2006, 08      & 12           \\ 
8.35~GHz      & 3.6e+04 &     1.3e+01 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     3.2e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     1.2e+01 & 1994         & 30           \\ 
3C~270        & & & & \\ 
Chandra      & 4.5e-04 &     6.6e-05 & $<$2005,01   & 1            \\ 
Chandra      & 2.0e-04 &     3.6e-05 & 2000, 03, 05   & 9            \\ 
F25SRF2      & 2.5e-01 & $<$2.1e-04 & 2000, 03, 05   & 9            \\ 
{\it F547M}        & 5.5e-01 &     6.0e-03 & 1994, 12, 13   & 7            \\ 
{\it F791W}        & 7.9e-01 &     1.1e-02 & 1994, 12, 13   & 7            \\ 
{\it F110W}        & 1.1e+00 &     2.8e-04 & 1998, 04, 23   & 6            \\ 
{\it F160W}        & 1.6e+00 &     1.0e-03 & 1998, 04, 23   & 6            \\ 
{\it F205W}        & 2.0e+00 &     2.2e-03 & 1998, 04, 23   & 6            \\ 
L            & 3.4e+00 &     1.2e+00 & 2000, 03, 21   & 31           \\ 
LW1          & 4.5e+00 & $<$7.0e+00 & 1996, 07, 02   & 31           \\ 
 \hline
LW2          & 6.7e+00 & $<$1.0e+01 & 1996, 07, 02   & 31           \\ 
VISIR/SiC    & 1.3e+01 & $<$2.3e+00 & 2006, 06, 18   & $ $           \\ 
LW3          & 1.4e+01 &     1.9e+01 & 1996, 07, 02   & 31           \\ 
345~GHz       & 8.7e+02 &     1.7e+02 & 2001, 02, 11   & 31           \\ 
43~GHz        & 7.0e+03 &     3.0e+02 & 1997, 09, 07   & 22           \\ 
22~GHz        & 1.4e+04 &     1.6e+02 & 1997, 09, 07   & 22           \\ 
8.4~GHz       & 3.6e+04 &     1.0e+02 & 1995, 04, 01   & 23           \\ 
5~GHz         & 6.0e+04 &     3.1e+02 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     2.4e+02 & 1994         & 30           \\ 
1.6~GHz       & 1.9e+05 &     1.0e+02 & 1995, 04, 01   & 23           \\ 
1.4~GHz       & 2.1e+05 &     1.4e+02 & 1993, 03      & 2            \\
Centaurus~A  & & & & \\ 
WFPC2        & 8.1e-01 &     7.0e-03 & 1997.80      & 27           \\ 
NACO/1.28    & 1.3e+00 &     1.3e+00 & 2003.45      & 28           \\ 
{\it F160W}        & 1.6e+00 &     4.8e+00 & 1997.69      & 27           \\ 
NACO/1.67    & 1.7e+00 &     4.5e+00 & 2003.45      & 28           \\ 
NACO/2.15    & 2.1e+00 &     3.4e+01 & 2004.25      & 28           \\ 
NICMOS/2.22  & 2.2e+00 &     4.2e+01 & 1997.61      & 27           \\ 
NACO/3.8     & 3.8e+00 &     2.0e+02 & 2003.36      & 28           \\ 
MIDI/8.3     & 8.3e+00 &     4.7e+02 & 2005.28      & 28           \\ 
MIDI/9.3     & 9.3e+00 &     2.8e+02 & 2005.28      & 28           \\ 
MIDI/10.4    & 1.0e+01 &     2.5e+02 & 2005.28      & 28           \\ 
MIDI/11.4    & 1.1e+01 &     4.3e+02 & 2005.28      & 28           \\ 
MIDI/12.6    & 1.3e+01 &     6.2e+02 & 2005.28      & 28           \\ 
VISIR/SiC    & 1.3e+01 &     1.2e+03 & 2006, 12, 27   & $ $           \\ 
667~GHz       & 4.5e+02 &     6.3e+03 & 1991.35      & 20           \\ 
375~GHz       & 8.0e+02 &     8.5e+03 & 1991.35      & 20           \\ 
270~GHz       & 1.1e+03 &     5.9e+03 & 2003.30      & 28           \\ 
235~GHz       & 1.3e+03 &     5.8e+03 & 2003.18      & 28           \\ 
150~GHz       & 2.0e+03 &     6.9e+03 & 2003.18      & 28           \\ 
90~GHz        & 3.5e+03 &     8.6e+03 & 2003.18      & 28           \\ 
22.2~GHz      & 1.4e+04 &     3.5e+03 & 1995.88      & 34           \\ 
22~GHz        & 1.4e+04 &     6.6e+03 & 2002,03      & 32           \\ 
18.5~GHz      & 1.6e+04 &     6.5e+03 & 2002,03      & 32           \\ 
8.4~GHz       & 3.6e+04 &     2.4e+03 & 1996.22      & 34           \\ 
5~GHz         & 6.0e+04 &     7.0e+03 & 1989         & 29           \\ 
4.8~GHz       & 6.3e+04 &     1.2e+03 & 1993.13      & 34           \\ 
2.3~GHz       & 1.3e+05 &     5.5e+03 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     5.7e+03 & 1993--2003    & 19           \\ 
PKS1417$-$19   & & & & \\ 
VISIR/SiC    & 1.3e+01 &     2.6e+01 & 2006, 12, 27   & $ $           \\ 
5~GHz         & 6.0e+04 &     5.2e+00 & $<$1993      & 37           \\ 
3C~317        & & & & \\ 
Chandra      & 4.5e-04 &     2.3e-05 & $<$2005, 01   & 1            \\ 
ROSAT/PSPC   & 1.2e-03 &     4.6e-05 & 2000, 09, 03   & 36           \\ 
{\it F210M}        & 2.2e-01 &     7.7e-04 & 1994, 03, 05   & 9            \\ 
{\it F555W}       & 5.6e-01 &     1.5e-02 & 1997, 10, 18   & 7            \\ 
{\it F702W}        & 7.0e-01 &     2.0e-02 & 1994, 03, 05   & 9            \\ 
{\it F814W}        & 8.1e-01 &     2.1e-02 & 1997, 10, 18   & 7            \\ 
{\it F160W}        & 1.6e+00 &     1.2e-01 & 1998         & 36           \\ 
VISIR/SiC    & 1.3e+01 & $<$3.2e+00 & 2006, 06, 18   & $ $           \\ 
345~GHz       & 8.7e+02 &     9.1e+01 & 2001, 02, 11   & 31           \\ 
5~GHz         & 6.0e+04 &     3.9e+02 & 1989, 09      & 29           \\ 
2.3~GHz       & 1.3e+05 &     3.1e+02 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     5.0e+02 & $\dots$   & 2            \\ 
3C~327        & & & & \\ 
VISIR/SiC    & 1.3e+01 &     7.8e+01 & 2005, 07, 02   & $ $           \\ 
22.4~GHz      & 1.3e+04 &     1.1e+01 & 2006, 08      & 12           \\ 
14.9~GHz      & 2.0e+04 &     1.5e+01 & 2006, 08      & 12           \\ 
8.35~GHz      & 3.6e+04 &     2.5e+01 & 89, 91        & 17           \\ 
5~GHz         & 6.0e+04 &     2.8e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     3.2e+01 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     2.1e+01 & 1993, 03      & 2            \\ 
3C~353        & & & & \\ 
\hline 
F702W        & 7.0e-01 & $<$1.9e-03 & $<$1999, 4    & 8            \\ 
F160W        & 1.6e+00 & $<$2.0e-01 & 2004, 09, 09   & 26           \\ 
VISIR/SiC    & 1.3e+01 & $<$4.5e+00 & 2005, 07, 02   & $ $           \\ 
345~GHz       & 8.7e+02 &     7.4e+01 & 2001, 02, 11   & 31           \\ 
22~GHz        & 1.4e+04 &     1.9e+02 & 2002, 03      & 32           \\ 
18.5~GHz      & 1.6e+04 &     1.5e+02 & 2002, 03      & 32           \\ 
8.4~GHz       & 3.6e+04 &     1.5e+02 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     1.2e+02 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     7.8e+01 & 1994         & 30           \\ 
PKS1839$-$48   & & & & \\ 
VISIR/SiC    & 1.3e+01 & $<$4.1e+00 & 2006,06,18   & $ $           \\ 
24~GHz        & 1.2e+04 &     1.0e+02 & 2006,07      & 12           \\ 
22~GHz        & 1.4e+04 &     9.3e+01 & 2002,03      & 32           \\ 
18.5~GHz      & 1.6e+04 &     2.0e+02 & 2002,03      & 32           \\ 
18~GHz        & 1.7e+04 &     1.1e+02 & 2006,07      & 12           \\ 
5~GHz         & 6.0e+04 &     1.6e+02 & 1991-1992    & 29           \\ 
2.3~GHz       & 1.3e+05 &     1.3e+02 & 1994         & 30           \\ 
3C~403        & & & & \\ 
{\it F702W}        & 7.0e-01 &     6.5e-03 & $<$1999,04   & 8            \\ 
{\it F160W}        & 1.6e+00 & $<$7.9e-02 & 2004, 11, 06   & 26           \\ 
VISIR/SiC    & 1.3e+01 &     8.9e+01 & 2006, 06, 18   & $ $           \\ 
22.4~GHz      & 1.3e+04 &     8.5e+00 & 2006, 08      & 12           \\ 
14.9~GHz      & 2.0e+04 &     2.0e+01 & 2006, 08      & 12           \\ 
8.35~GHz      & 3.6e+04 &     7.1e+00 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     1.0e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     2.4e+01 & 1994         & 30           \\ 
3C~403.1      & & & & \\ 
VISIR/SiC    & 1.3e+01 & $<$4.2e+00 & 2006, 06, 18   & $ $           \\ 
3C~424        & & & & \\ 
F28X50LP     & 7.0e-01 & $<$1.5e-02 & 2000, 03, 15   & 9            \\ 
VISIR/SiC    & 1.3e+01 & $<$3.7e+00 & 2006, 06, 18   & $ $           \\ 
8.4~GHz       & 3.6e+04 &     7.0e+00 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     1.8e+01 & $ $   & 3            \\ 
PKS2158$-$380  & & & & \\ 
F130M        & 1.3e+00 &     2.2e-02 & 1991, 05, 07   & 5            \\ 
VISIR/SiC    & 1.3e+01 &     1.7e+01 & 2006, 06, 18   & $ $           \\ 
5~GHz         & 6.0e+04 & $<$7.0e+00 & 1982         & 15           \\ 
3C~445        & & & & \\ 
{\it F702W}        & 7.0e-01 &     2.8e+00 & $<$1999, 04   & 8            \\ 
{\it F160W}        & 1.6e+00 & $<$ 2.1e-01 & 2005, 06, 24   & 26           \\ 
VISIR/SiC    & 1.3e+01 &     1.9e+02 & 2005, 07, 02   & $ $           \\ 
22.4~GHz      & 1.3e+04 &     1.6e+01 & 2006, 08      & 12           \\ 
22~GHz        & 1.4e+04 &     4.5e+01 & 2002, 03      & 32           \\ 
18.5~GHz      & 1.6e+04 &     3.9e+01 & 2002, 03      & 32           \\ 
14.9~GHz      & 2.0e+04 &     2.7e+01 & 2006, 08      & 12           \\ 
8.4~GHz       & 3.6e+04 &     8.4e+01 & 1989--1991    & 17           \\ 
5~GHz         & 6.0e+04 &     8.6e+01 & 1989         & 29           \\ 
2.3~GHz       & 1.3e+05 &     9.0e+01 & 1994         & 30           \\ 
1.4~GHz       & 2.1e+05 &     6.9e+01 & 1993--2003    & 2            \\ 
PKS2354$-$35   & & & & \\ 
VISIR/SiC    & 1.3e+01 & $<$3.1e+00 & 2006, 06, 18   & $ $           \\ 
5~GHz         & 6.0e+04 &     5.0e+00 & $<$1981      & 13           \\ 
\hline
\end{tabular}
\par
%\tablefoot {\textit{Column~5)} References from:
\tablebib{ 
1~=~\cite{2006AetA...451...35B}, 2~=~\cite{2003yCat.8071....0B}, 3~=~\cite{1992MNRAS.256..186B}, 4~=~\cite{1994AetAS..105...91B}, 5~=~\cite{1996AetA...305..715B}, 6~=~\cite{2000MNRAS.318..493C}, 7~=~\cite{1999AetA...349...77C}, 8~=~\cite{2000AetA...355..873C}, 9~=~\cite{2002AetA...394..791C}, 10~=~\cite{2002ApJ...571..247C}, 11~=~\cite{2003ApJ...582..645C}, 12~=~\cite{2008ApJ...678..712D}, 13~=~\cite{1989MNRAS.236..737E}, 14~=~\cite{1994ApJ...434...67F}, 15~=~\cite{1982MNRAS.201..991F}, 16~=~\cite{1988AetA...199...73G}, 17~=~\cite{1998MNRAS.296..445H}, 18~=~\cite{2003MNRAS.338..176H}, 19~=~\cite{2007ApJ...670L..81H}, 20~=~\cite{1993MNRAS.260..844H}, 21~=~\cite{2006MNRAS.368..609J}, 22~=~\cite{2000ApJ...534..165J}, 23~=~\cite{1997ApJ...484..186J}, 24~=~\cite{2005AJ....129.2138L}, 25~=~\cite{1991AJ....102..537L}, 26~=~\cite{2006ApJS..164..307M}, 27~=~\cite{2000ApJ...528..276M}, 28~=~\cite{2007AetA...471..453M}, 29~=~\cite{1993MNRAS.263.1023M}, 30~=~\cite{1997MNRAS.284..541M}, 31~=~\cite{2003AJ....126.2677Q}, 32~=~\cite{2006AetA...445..465R}, 33~=~\cite{1986MNRAS.219..545S}, 34~=~\cite{1998AJ....115..960T}, 35~=~\cite{1999AetA...348..437T}, 36~=~\cite{2003AetA...403..889T} and 37~=~\cite{1995ApJ...448..521Z}.
}
\end{table}